Author's response to reviews

Title: Homeopathic Medical Practice: Long-term results of a Cohort Study with 3981 Patients

Authors:

Claudia Witt (claudia.witt@charite.de)
Rainer Ludtke (r.luedtke@carstens-stiftung.de)
Roland Baur (claudia.witt@charite.de)
Stefan N Willich (stefan.willich@charite.de)

Version: 2 Date: 29 August 2005

Author's response to reviews: see over
Response to Reviewer

The reference of Hannan 1994 was added in the reference section (page 16 reference 16)

We specified the sentence that the study was the first to evaluate classical homeopathy (see page 3 para 2 and page 10 para 6).

We discussed the problem of measuring quality of life in children more in detail (see page 11, para 5).

In Figure 2 we added the detailed percentages next to the bars.

The statistic section was enlarged by explaining the used regression models more in detail (see page 6 and 7).